Contents lists available at ScienceDirect





# Journal of Economic Behavior and Organization

journal homepage: www.elsevier.com/locate/jebo

# Information about inequality in early child care reduces polarization in policy preferences $\stackrel{\text{\tiny{$\bigstar$}}}{=}$

Henning Hermes<sup>a</sup>, Philipp Lergetporer<sup>b,a</sup>, Fabian Mierisch<sup>c,\*</sup>, Guido Schwerdt<sup>d</sup>, Simon Wiederhold<sup>e,a,f</sup>

<sup>a</sup> ifo Institute, Germany

<sup>b</sup> Technical University of Munich, TUM School of Management, Campus Heilbronn, Germany

<sup>c</sup> Catholic University of Eichstaett-Ingolstadt, Germany

<sup>d</sup> University of Konstanz, Germany

e Halle Institute for Economic Research (IWH) and University of Halle, Germany

<sup>f</sup> Hoover Institution, Stanford University, United States of America

# ARTICLE INFO

JEL classification: 124 118 J13 D83 C99 Keywords: Child care Policy support Information Inequality Discrimination Survey experiment

# ABSTRACT

We investigate public preferences for equity-enhancing policies in access to early child care, using a survey experiment with a representative sample of the German population ( $n \approx 4,800$ ). We observe strong misperceptions about migrant-native inequalities in early child care that vary by respondents' age and right-wing voting preferences. Randomly providing information about the actual extent of inequalities has a nuanced impact on the support for equity-enhancing policy reforms: it increases support for respondents who initially underestimated these inequalities, and tends to decrease support for those who initially overestimated them. This asymmetric effect leads to a more consensual policy view, substantially decreasing the polarization in policy support between under- and overestimators. Our results suggest that correcting misperceptions can align public policy preferences, potentially leading to less polarized debates about how to address inequalities and discrimination.

# 1. Introduction

Inequality of opportunity is a pressing societal challenge in many countries around the world. The opportunities available to individuals are frequently shaped within the education system (Corak, 2013), where socioeconomic disparities in accessing quality education emerge at an early stage. These inequalities are already evident in early child care, where disadvantaged groups are strongly underrepresented (OECD, 2018; García et al., 2020; Heckman and Landersø, 2021).<sup>1</sup> This situation is particularly concerning

Corresponding author.

# https://doi.org/10.1016/j.jebo.2024.106780

Received 25 January 2024; Received in revised form 5 October 2024; Accepted 12 October 2024

Available online 1 November 2024

 $<sup>\</sup>stackrel{i}{\sim}$  We thank Alexander Cappelen, Bertil Tungodden, as well as the conference and seminar participants at the Centre for Experimental Research on Fairness, Inequality and Rationality (FAIR), the Workshop on Field Experiments in Economics and Business, the Advances with Field Experiments in Chicago, the Bavarian Micro Day, the University of Konstanz, and the Catholic University of Eichstaett-Ingolstadt for their helpful comments. This project is funded by the Deutsche Forschungsgemeinschaft (DFG – German Research Foundation) under Germany's Excellence Strategy -EXC-2035/1, 390681379. The study was pre-registered in the AEA RCT Registry (AEARCTR-0010455). During the preparation of this work the authors used ChatGPT in order to improve the writing. After using this tool, the authors reviewed and edited the text as needed and take full responsibility for the content of the publication.

E-mail address: Fabian\_Mierisch@gmx.de (F. Mierisch).

<sup>&</sup>lt;sup>1</sup> Socioeconomic disparities in access to early child care can emerge for several reasons, including complex admission processes (Hermes et al., 2024) and discriminatory behavior of child care center managers (Hermes et al., 2023a). Across most OECD countries, the resulting socioeconomic enrollment gaps are substantial, even when accounting for parental enrollment preferences (OECD, 2018; Jessen et al., 2020).

<sup>0167-2681/© 2024</sup> The Authors. Published by Elsevier B.V. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

given the pivotal role of early child care in fostering equality of opportunity and enabling disadvantaged groups to actively engage and contribute to various societal sectors (e.g., Almond and Currie, 2011; Heckman et al., 2013; Campbell et al., 2014). Addressing these foundational inequalities in early childhood is therefore a crucial policy imperative.

However, the implementation of policies that enhance equity is at risk of being compromised by the increasingly polarized political discourse concerning policies aimed at disadvantaged groups, such as minorities or migrants (Dixit and Weibull, 2007; Karakas and Mitra, 2019; Bonomi et al., 2021; Guriev and Papaioannou, 2022). This polarization could potentially deadlock initiatives designed to tackle societal inequality. Many suggest that a primary cause of political disagreement over policies promoting equity – such as anti-discrimination measures, affirmative action, or transparency requirements – stems from differing perceptions of the extent of these inequalities (Settele, 2022; Bursztyn and Yang, 2022; Haaland and Roth, 2023). However, public perceptions about inequality in access to early child care have not yet been investigated.

This study aims to understand the public's perception of migrant-native inequalities in the early child care market, as well as the level of support for equity-enhancing policy reforms. Studying public policy preferences, as we do, has become increasingly popular in economic research in recent years (Haaland et al., 2023), given their crucial role in determining the political viability of reform proposals. Public policy preferences not only directly influence politicians' policy stances (Hager and Hilbig, 2020) but also predict real-world political behaviors, such as signing petitions (see Section 6 for details).<sup>2</sup> Most importantly, we employ an information provision experiment (Haaland et al., 2023) to causally examine how people's perceptions of these gaps affect their preferences for policy reforms. We investigate the interplay between prior beliefs and reform support, and the information effect on polarization of policy preferences. We implemented the experiment in a large-scale survey with about 4800 respondents representative of the German population in terms of gender, age, educational background, and residential state.

In the survey, we first elicit respondents' beliefs regarding migrant-native gaps in the early child care market. Our study reveals two important descriptive findings regarding the general public's beliefs. First, we observe substantial variation in prior beliefs about migrant-native gaps, indicating a general lack of knowledge (or awareness) about this issue. Second, respondents consistently underestimate existing inequalities in child care enrollment rates. At the same time, they overestimate the migrant-native gap in response rates to child-care-related email inquiries by parents.<sup>3</sup> Investigating the correlation of prior beliefs about migrant-native inequalities with observable characteristics, we find that respondents who are younger, more educated, and have non-right-wing voting preferences, respectively, are less likely to underestimate inequalities in the child care market.

We also elicited support for various equity-enhancing policy reforms: the provision of (publicly subsidized) additional slots, a centralized admission process, additional financial incentives for child care centers to admit migrants, and preferential admission of migrants. These policies were chosen as existing literature suggests their effectiveness in addressing various dimensions of unequal access to early child care (see Section 3.2 for details). The provision of additional slots was the most favored, garnering support by 70% of survey respondents in the control group. A centralized admission process followed with 40% support. Financial incentives and preferential treatment of migrants were less popular, receiving only 25% and 7% support, respectively. We further observe a notable correlation between prior beliefs about migrant-native gaps (specifically regarding child care centers' responsiveness to emails) and support for equity-enhancing policy reforms. To determine if this correlation indicates a causal relationship, we experimentally analyze the impact of exogenous shifts in these beliefs on reform support.

In the experiment, we randomly provide information about the actual extent of migrant-native disparities in early child care. Studying the causal relationship between beliefs and policy preferences with observational data is challenging due to the lack of exogenous variation in beliefs about migrant-native inequalities, and a lack of individual-level data on reform preferences. Our experimental survey addresses these identification challenges. The first treatment informs about the migrant-native *enrollment* gap in early child care (Jessen et al., 2020). The second treatment informs about the migrant-native gap in *response rates* of child care centers to parental email requests (Hermes et al., 2023a), and the third treatment combines both pieces of information. In the treatments, information is conveyed both verbally and through simple visual representations. The control group does not receive any of this information.

Providing information about the extent of migrant-native inequalities has no significant effect on reform support on average. However, the information treatments are successful in updating respondents' beliefs. Scrutinizing respondents' perception of the reasons for unequal chances between migrants and natives, 27% of control-group respondents attribute it to the cultural background of migrants.<sup>4</sup> Providing treatment information significantly increases the perception that disparities are based on cultural background by 3.8 percentage points (13.8%, p = .017).

It is reasonable to expect that the extent to which information provision affects reform support crucially depends on respondents' prior beliefs about existing inequalities. Therefore, we investigate how the reaction to treatment information depends on prior beliefs.

<sup>&</sup>lt;sup>2</sup> In Germany, public political engagement, such as protests and referenda on issues like child care policies (SZ, 2024), school policies (Spiegel, 2010), and university tuition (Lergetporer and Woessmann, 2023), occurs regularly and sometimes directly affects education policy making.

<sup>&</sup>lt;sup>3</sup> We focus on Turkish migrants, as they represent the largest migrant group in Germany (see Section 3 for details). Respondents, on average, believe that 36.5 out of 100 Turkish migrant children are enrolled in early child care, while this is true for only 12 out of 100 (Jessen et al., 2020). The migrant-native gap is 21 percentage points, as 33 out of 100 native children are enrolled. Respondents also believe that, on average, 52.4 out of 100 parents of Turkish migrant children receive a response to an inquiry, while this is true for 63 out of 100 (Hermes et al., 2023a). As 71 out of 100 natives receive a response, the migrant-native gap here is 8 percentage points. Crucially, the questions that elicited respondents' beliefs about migrants' enrollment and email response rates included information on the corresponding figures for natives, an anchoring technique commonly employed to reduce measurement error and clarify the interpretation of possible information treatment effects (see Section 3.1).

<sup>&</sup>lt;sup>4</sup> More effort required from child care centers to cater to migrant children (51%) and preferences of other parents (46%) were regarded as even more relevant by respondents. Note that multiple answers could be selected.

Respondents who *underestimate* inequalities in the child care market (i.e., respondents who initially underestimated migrant-native gaps) exhibit relatively little support for equity-enhancing policies, while those *overestimating* these gaps are more in favor of such policies. However, when respondents who underestimate inequalities receive information about the actual gaps, they significantly increase their reform support. Those who initially overestimate the gaps tend to decrease reform support upon receiving information, albeit not statistically significantly so. As a result, information provision leads to more consensual reform preferences: the gap in reform support between underestimators and overestimators decreases by 43% in the treatment group as compared to the control group. We confirm this finding in a Causal Forest analysis, which shows that treatment effect heterogeneities depend strongly on prior beliefs about native-migrant gaps.

Furthermore, our large sample size allows us to explore treatment effect heterogeneities across various subgroups. We observe the pattern of converging reform support between females and males and between parents and non-parents.<sup>5</sup> However, one group shows a strikingly different pattern of belief updating: right-wing voters. Compared to other respondents, they generally view the child care market as less discriminatory against migrants and exhibit significantly lower support for equity-enhancing policy reforms. Interestingly, they tend to counter-intuitively *reduce* their policy support upon receiving accurate information. This reaction is consistent with an amplification of party identification discussed in the political-science literature, a mechanism to avoid the discomfort associated with challenging one's own beliefs (Campbell, 1980; Bartels, 2002).

Our study contributes to the existing literature in two key ways. First, we build upon prior research that examines policy preferences about the education system (e.g., Lergetporer et al., 2018; Cattaneo et al., 2020). Educational inequality early in life has profound consequences, paving the way for disparities in lifetime income and human capital accumulation later on (Heckman et al., 2010; Hermes et al., 2023a). However, our study is the first to investigate the causal determinants of public support for policy reforms aimed specifically at promoting equity in access to early child care. Doing so, we extend the work of Haaland and Roth (2023), who studied the effect of providing information about gaps in response rates to applications by white and black Americans on support for pro-black policies. We also explore the impacts of different types of information provided.

Second, our study contributes to the growing body of literature that investigates the impacts of information on the polarization of policy preferences. Previous survey experiments have predominantly shown that additional information about minorities leads to increased polarization of policy preferences (e.g., Naumann et al., 2018; Lergetporer et al., 2021; Settele, 2022), or has little effect on polarization (e.g., Hopkins et al., 2019; Alesina et al., 2023; Haaland and Roth, 2023). In contrast, our study provides evidence suggesting that information about minorities can actually decrease polarization, thereby facilitating the formation of more consensual policy reform preferences. Notably, in a recent literature survey on information provision experiments, Marino et al. (2024) identify only one study in which information about the share of (undocumented) immigrants reduces polarization in immigration policy preferences (Grigorieff et al., 2020). A potential explanation for our findings is the wide variation in respondents' prior beliefs, suggesting a general lack of awareness or knowledge about migrant-native inequalities in early child care. Consequently, these beliefs may be more readily updated upon exposure to new information, as they are less entangled with respondents' identity concerns.

The remainder of this paper is structured as follows. Section 2 provides information on the institutional background of the early child care market in Germany. Section 3 describes the survey data and the experimental design. Section 4 reports our results regarding respondents' prior beliefs about early child care for migrants and average treatment effects on support for policy reforms. Section 5 analyzes treatment effect heterogeneities. Section 6 provides a discussion on the potential limitations of our study and an outlook on how future research may address them. Section 7 concludes.

#### 2. Institutional background

In Germany, child care is available to all children up until they begin school at the age of six, with specific provision for two age groups: (i) children under the age of three years (*"Krippe"*), and (ii) children between the ages of three and six years (*"Kindergarten"*). Every child is entitled to a child care slot from the age of one year onward. The government subsidizes early child care, covering approximately three-quarters of the total cost (Spiess, 2013). Parents pay very low child care fees (on average 250 EUR per month), equivalent to 10% of the average income. Lower-income families are eligible for fee reductions or exemptions (Felfe and Lalive, 2018). Compared to other countries, the quality of early child care in Germany is relatively high and homogeneous, as measured by group sizes, staff-to-child ratios, and other indicators (Felfe and Lalive, 2018).

While child care in Germany is often described as "universal", the reality is quite different. For instance, only about 34% of children under three years of age are enrolled in early child care. This figure increases significantly for children between two and three years, with a 55% enrollment rate. Notably, over 90% of children attend Kindergarten, indicating widespread participation in some form of child care prior to school. These statistics, as reported in Education Report (2020), highlight a shift in focus from mere access to child care to the specifics of timing and early enrollment. Supporting the relevance of the timing of child care, previous research has demonstrated that early enrollment in child care can significantly enhance a child's development (Drange and Havnes, 2019).

Part of the reason for the relatively low enrollment rates in early child care is the shortage of available child care slots, leading to widespread rationing. Importantly, this issue disproportionately affects parents with a migration background. Although the wish to

<sup>&</sup>lt;sup>5</sup> Females, initially with lower reform support, show a greater increase than males when provided with information. Conversely, parents, initially more supportive of reforms, exhibit a more pronounced decrease in support upon receiving information.

enroll children in early child care is similar among both native and migrant parents, there is a notable disparity in actual enrollment rates. For instance, only 21% of children with a migration background are enrolled in early child care, compared to 33% of native children (Jessen et al., 2020). This indicates a significant gap in child care access between migrants and natives.

Child care in Germany falls under the purview of the child and youth welfare system, with the federal government bearing responsibility. Nonetheless, the actual provision of child care is managed at the municipality level. A significant majority of child care centers, approximately 83%, are operated by municipalities, non-profit organizations, and associations. In contrast, private for-profit providers constitute a mere 3% of all child care facilities (see Education Report, 2020). The provision of child care services is primarily carried out by small centers, typically catering to 25–75 children (DJI, 2021). Competition between child care centers is generally low (Spiess, 2013).

The German child care market is characterized by a decentralized structure, with each municipality – and often each center – having its own distinct enrollment process. As a result, the allocation process of child care slots is often criticized as very complicated, non-transparent, and inefficient. Families often face divergent experiences: while some wait years to secure a slot, despite their legal entitlement, others receive multiple offers, inadvertently blocking access for others and prolonging waiting times. The absence of mandatory, standardized criteria for slot allocation and a lack of a centralized system to monitor enrollment decisions exacerbate the difficulties in navigating the application process. The decentralized nature of child care admissions creates conditions that are potentially conducive to high inequality and discrimination (Hermes et al., 2024, 2023a,b).

#### 3. Data and experimental design

## 3.1. The survey

We implemented our experiment in the second wave of the *Inequality Barometer* of the University of Konstanz. The online survey was conducted in late November 2022, and aimed to capture public perceptions of inequality. The survey was conducted by the survey company Verian (previously known as Kantar Public), and consisted of seven modules, with our experiment being the fourth. The sample includes 4822 respondents drawn using quotas to represent the German voting-age population (18 years and older) in terms of gender, age, state of residence, and education background. In addition, the survey company provides survey weights to adjust for minor deviations of the sample from the general population, which we applied in all empirical analyses.

The survey was conducted using Verian's Payback Online Panel, a non-probability sample of approximately 150,000 active members. Recruitment takes place through the Payback loyalty program, one of the largest consumer bonus programs in Germany with around 25 million consumers, representing roughly half of all German households. While initial recruitment into the panel may involve non-probability elements, the sampling process from the panel aligns with principles of probability sampling, within the constraints of an online access panel. Overall, about 40% of invited panel members completed the survey, around 5% dropped out, and another 5% were screened out based on quotas, indicating a take-up rate (clicking on the invitation) of about 50%. Median completion time for the full survey was 20 min. As an incentive to participate, respondents received 2 EUR in the form of credit points for a voucher system. Further information about the survey and the sampling procedure, as well as screenshots of the main questions, can be found in Appendix C.

The objective of this study is to evaluate existing beliefs about migrant-native gaps in early child care, and investigate the impact of providing information about these inequalities on public preferences for equity-enhancing reforms. In our experiment, we randomly assign respondents to different experimental groups which receive different pieces of information about inequality in early child care before stating their reform support.

The survey module begins by eliciting respondents' initial beliefs regarding migrants' enrollment rate, and child care centers' response rate to email inquiries from migrant parents. Respondents state these beliefs only for migrant parents, while we provide the correct rates for natives as reference points. We focus on Turkish migrants, who represent the largest and geographically most dispersed migrant group in Germany. In 2019, there were approximately 1.5 million people of Turkish origin in Germany, accounting for approximately 1.3% of the German population and 13% of all migrants in Germany (Bundesamt für Migration, 2019). Furthermore, Turkish migrants are severely underrepresented in early child care, as their enrollment rate is 21 percentage points below the rate of natives, while demand for child care is very similar in both groups (Jessen et al., 2020).

In particular, we ask respondents to answer the following questions: (i) "Please give your assessment of Turkish parents. How many out of 100 children of Turkish parents attend a daycare center (for children under the age of 3)? For your orientation, we provide you with the figures for German parents. According to a scientific study, 33 out of 100 children of German parents attend a daycare center". (ii) "Please give your assessment of Turkish parents. How many Turkish parents who send an e-mail inquiry to a daycare center receive a reply? According to a scientific study, 71 out of 100 German parents receive a reply to an email inquiry from a child care center. ..." Respondents answer using a slider to indicate a numerical value ranging from zero to 100 (see Appendix C.2 for screenshots).

Note that our belief-elicitation questions anchored respondents' beliefs about natives' enrollment and email response rates when eliciting beliefs about migrants. Such anchoring is commonly employed to reduce measurement errors in elicited beliefs (Roth et al., 2022; Lergetporer and Woessmann, 2023). In our study, the anchors are crucial for clarifying the interpretation of possible information treatment effects; without an anchor, any observed treatment effects could potentially arise from updating beliefs about either natives' or migrants' rates. The anchors allow us to attribute any information effects specifically to updates regarding migrant figures (or the migrant-native gap).

# 3.2. Experimental design

After eliciting prior beliefs, respondents are randomly assigned to one of three treatment groups, or the control group. Each treatment group receives a specific piece of information, including a graphical representation, as shown in Appendix C.3. We provide the following treatment information:

- (i) "T1: Enrollment rate information": Respondents receive information comparing the enrollment rate of children from German parents (33/100) with that of children from Turkish parents (12/100) in early child care (Jessen et al., 2020).
- (ii) "T2: Response rate information": Respondents receive information about the response rate of child care centers to inquiries from German parents (71/100) compared to inquiries from Turkish parents (63/100) (Hermes et al., 2023a).
- (iii) "T3: Enrollment & response rate information": Respondents receive both sets of information, i.e., the combination of T1 and T2.

In all treatments we display the source of the information with a citation.<sup>6</sup> We also include a mouse-over text box offering a concise summary of the studies, accessible when respondents hover over an information icon, to assure respondents that the provided information is evidence-based. Note that our treatment design adheres to best practices in designing information provision experiments (Haaland et al., 2023), such as using official statistics and enhancing verbal statements with graphical depictions of the information.

We then measure respondents' support for four equity-enhancing policy reforms on a five-point Likert scale ranging from "I fully disagree" to "I fully agree". The four policy reforms are (i) introducing centralized child care admission processes at the municipal level, (ii) providing additional child care slots, (iii) implementing preferential treatment of migrant families in the enrollment process, and (iv) offering child care centers additional incentives to admit migrant children (see Appendix Figure C5 for the original presentation and wording). Based on the existing body of literature, we selected policy interventions deemed likely to mitigate the migrant-native disparities in early child care access. The initial two (non-targeted) proposals aim to alleviate slot shortages, either through the direct provision of additional slots or by enhancing efficiency in slot distribution via centralization, which is regarded as a key strategy for fostering equitable access to child care (Jessen et al., 2020).<sup>7</sup> However, emerging evidence indicates that non-targeted policies alone are insufficient to completely bridge the gap for migrant families, who may struggle with the child care application process (Hermes et al., 2024) or face discrimination from child care center administrators (Hermes et al., 2023a). The two policies specifically aimed at migrants directly address these barriers to equitable child care access.

To reduce potential biases from imperfect memory of the treatment information, we present a reminder to treated respondents in a text box on the screen where they indicate their policy reform preferences (see Appendix Figure C6 for an example).<sup>8</sup>

Furthermore, we elicit respondents' perceptions about the reasons behind migrants' disadvantages in the early child care market. To do so, respondents are given the option to select multiple reasons from the following list: Unequal treatment due to (i) the migrants' cultural background, (ii) the additional effort required from child care centers to cater to migrants, and (iii) the preferences of other parents. Additionally, respondents could select the options "other reasons", "don't know", and "not specified" (see Appendix Figure C7). We use these perceptions as an indirect manipulation check to assess whether the treatment influences respondents' perceptions. Although a more direct test would involve repeating the prior-belief questions post-treatment, we avoided this to prevent confusion among participants, particularly in the control group, which could result in biased response behavior (see, e.g., Grewenig et al., 2020).

#### 3.3. Econometric model

We estimate the treatment effects using OLS regressions of the specific outcome of interest on randomized treatment indicators. Our main specification is the following:

$$Y_i = \alpha_1 + \beta_1 T \mathbf{1}_i + \beta_2 T \mathbf{2}_i + \beta_3 T \mathbf{3}_i + \mathbf{X}'_i \mathbf{\mu} + \epsilon_i$$
(1)

We define  $Y_i$  as the outcome of interest, e.g., reform support, for survey respondent *i*. To facilitate the interpretation of treatment effects on the overall support for policy reforms, we construct an index following Kling et al. (2007). In particular, we first z-standardize the support for each policy in the control group. Then, we calculate the mean of the four standardized policy support measures for each respondent, and z-standardize it again. The resulting composite measure captures overall treatment effects on reform support across multiple categories.

 $T1_i$ ,  $T2_i$ , and  $T3_i$  are binary indicators that take a value of one if respondent *i* received treatment 1 ("T1: Enrollment rate informaton"), treatment 2 ("T2: Response rate information"), or treatment 3 ("T3: Enrollment & response rate information"), and zero otherwise. Additionally, we construct an indicator variable called "Treatments (T1 | T2 | T3)" that takes a value of one if the respondent is assigned to any of the three treatment groups, and zero otherwise. This indicator allows us to examine the overall effect of being exposed to any information on a given outcome.

<sup>&</sup>lt;sup>6</sup> We also conducted another treatment which informed respondents about German and Turkish parents' enrollment wish. However, we have excluded this treatment from our main analysis due to its interpretational ambiguity. For details, see Appendix E.

<sup>&</sup>lt;sup>7</sup> The (decentralized) process of allocating child care slots frequently faces criticism for inefficiency. Some families endure years on waiting lists before securing a slot (Carlsson and Thomsen, 2015), while others receive multiple offers, thereby blocking access and prolonging wait times for other families (Fugger et al., 2017). A centralized admission system could reduce these inefficiencies.

<sup>&</sup>lt;sup>8</sup> Like many survey experiments in economics, our main outcomes of interest are survey-based stated policy preferences. These outcomes are sometimes criticized for lacking immediate economic or political consequences. In Section 6, we discuss several pieces of evidence highlighting the relevance of stated policy preferences for real-world political processes.

#### Table 1

Descriptive statistics and balancing

(2)

Variable	(1)			(2) T1: Enrollment rate information		(3) T2: Response rate information		(4) T3: Enrollment & response rate information				
	Control											
	N	Mean	SD	N	Diff.	P-value	N	Diff.	P-value	N	Diff.	P-value
Female	1223	0.513	(0.503)	1174	-0.008	0.683	1223	0.032	0.117	1202	0.001	0.975
Migrant	1211	0.159	(0.366)	1167	0.009	0.572	1206	-0.014	0.329	1192	-0.004	0.779
Parent	1223	0.418	(0.493)	1174	0.016	0.436	1223	0.017	0.391	1202	-0.013	0.526
Property owner	1171	0.454	(0.498)	1136	-0.023	0.267	1178	-0.021	0.297	1166	-0.008	0.685
Right-wing voter	1223	0.119	(0.324)	1174	0.009	0.549	1223	0.017	0.247	1202	-0.006	0.650
Age												
18–39 years	1223	0.298	(0.458)	1174	-0.000	0.986	1223	0.006	0.758	1202	0.011	0.555
40-59 years	1223	0.343	(0.475)	1174	0.012	0.546	1223	0.018	0.352	1202	-0.005	0.803
At least 60 years	1223	0.358	(0.480)	1174	-0.011	0.557	1223	-0.024	0.218	1202	-0.006	0.749
Education												
Lower education	1223	0.327	(0.469)	1174	-0.022	0.244	1223	-0.021	0.259	1202	-0.003	0.891
Medium education	1223	0.298	(0.457)	1174	0.018	0.329	1223	0.036*	0.056	1202	0.032*	0.091
Higher education	1223	0.375	(0.484)	1174	0.004	0.850	1223	-0.015	0.451	1202	-0.029	0.134
Prior beliefs												
Prior enrollment rate	1157	35.399	(25.900)	1111	1.385	0.208	1155	1.982*	0.070	1147	1.264	0.250
Prior response rate	1143	52.773	(29.104)	1095	-0.941	0.443	1142	0.263	0.828	1140	-0.676	0.579

Notes: Table shows means and standard deviations of variables for the control group. *Diff* reports the difference in means of the respective variable between the control group and each of the three treatment groups. We indicate the results of a two-sided t-tests between the control mean and the mean of each respective treatment group with significance stars. *Female:* Categorical variable taking a value of one if the respondent states to be female, two if the respondent states to be diverse, zero otherwise (the diverse category in the gender variable (n = 10) is not shown). *Migrant:* Indicator variable taking a value of one if the respondent states to be female, two if the respondent states to be diverse, zero otherwise (the diverse category in the gender variable (n = 10) is not shown). *Migrant:* Indicator variable taking a value of one if the respondent is a parent (has a least one child under the age of 18 in the household), zero otherwise. *Property owner:* Indicator variable taking a value of one if the respondent owns a house, zero otherwise. *Right-wing voter:* Indicator variable taking a value of one if the respondent stated to vote for a right-wing party (AfD, NPD, Dritter Weg, or Die Rechte), zero otherwise. *Age* "18–39 years" respondent is between 18 and 39 years old; "40–59 years" respondent is between 40 and 59 years old; "At least 60 years" respondent is 60 years and older. *Education:* "Higher education:" college entrance qualification, "Abitur"; "Medium education:" middle-tier secondary education ("Hauptschulabschluss"). *Prior beliefs:* Respondents' estimation how likely migrants enroll their child into child care and receive a response to child-care-related email inquiries, respectively (in percent). See Appendix D for detailed variable descriptions. Significance levels: \* p < .10, \*\* p < .05, \*\*\* p < .01.

Due to the randomized experimental design, the causal effect of information provision on the respective outcomes can be calculated from raw differences between treatment and control groups. However, we include a vector of control variables  $X_i$  for precision and to account for potential small imbalances across experimental groups. These controls comprise gender, age (respondent is 18 to 39 years, 40 to 59 years, or at least 60 years old), education (respondent's highest degree is secondary, upper secondary, or post-secondary education), and wealth status (respondent owns real estate or not).  $\epsilon_i$  is the idiosyncratic error term. We employ survey weights provided by the survey company throughout to align the drawn sample to known population counts.

We investigate potential treatment effect heterogeneities using the following model:

$$Y_i = \alpha_2 + \gamma_1 Treatments (T1|T2|T3)_i$$

+ 
$$\gamma_2 T$$
 reatments  $(T1|T2|T3)_i \times Subgroup$ 

+  $\gamma_3 Subgroup_i + \mathbf{X}'_i \mathbf{\delta} + v_i$ 

*Treatments*  $(T1|T2|T3)_i$  is an indicator that takes a value of one if respondent *i* is assigned to any of the three treatment groups, and zero otherwise. *Subgroup*<sub>i</sub> is an indicator that takes a value one if respondent *i* is part of a specific subgroup, and zero otherwise.

Since we expect the information treatment to operate through updating respondents' prior beliefs about migrant-native gaps in early child care, we are particularly interested in exploring heterogeneous treatment effects based on these prior beliefs. In our preferred specification, we divide individuals into two distinct groups. The first group ("underestimator" of inequalities) consists of individuals who consistently hold higher beliefs about migrants' enrollment and response rates compared to the actual rates (12 out of 100 and 63 out of 100, respectively). The second group ("overestimator") comprises all other individuals. Doing so, we examine whether treatment effects differ between individuals who under- or overestimate migrant-native gaps in early child care relative to the information presented in the treatments. We also explore treatment effect heterogeneities by sociodemographic subgroups.

# 3.4. Balancing

Table 1 presents the means and standard deviations of various respondent characteristics in the treatment groups compared to the control group. Overall, demographic characteristics are well balanced across the experimental groups. Three out of 39 pairwise comparisons turn out statistically significant at the 10%-level, which we would expect by pure chance. None of the differences is significant at the 5%-level.

In Appendix Table B1, we further assess the random assignment of treatments. We regress the treatment indicators on a set of control variables, including a dummy for item non-response and respondents' prior beliefs. The resulting F-statistics reject the joint



(b) Cumulative Distribution Functions of Prior Beliefs

Fig. 1. Prior beliefs about early child care for migrants. *Notes*: Subfigure (a) shows the mean answers of respondents to the prior belief elicitation questions for the enrollment rate of migrant children and the response rate to inquiries by migrant parents. Error bars indicate 95% confidence intervals. Subfigure (b) shows the corresponding cumulative distribution functions (CDFs). Both graphs also depict the actual value for migrants (*Actual*) as well as the values for natives that we provided to the respondents (*Anchor*).

significance of the explanatory variables (F = .46, F = 1.27, F = .87, F = .83, and F = .46 for Columns (1) to (5), respectively). This finding provides additional support for the conclusion that random assignment was successful.

# 4. Results on prior beliefs and average treatment effects

## 4.1. Descriptive findings

First, we present descriptive results regarding respondents' prior beliefs. On average, respondents believe that 35.4 out of 100 migrant children are enrolled in early child care (see Fig. 1(a)). This value significantly overestimates the actual enrollment rate of just 12 out of 100 migrants. Interestingly, respondents believe that migrants' enrollment rate is slightly higher than the one of natives (33 out of 100; see anchor in Fig. 1(a)), while in reality migrants are strongly underrepresented (see Jessen et al., 2020).

Turning to beliefs about child care centers' responses to parental email inquiries, respondents believe that migrant parents receive responses to 52.4 out of 100 emails inquiries. However, the actual value is higher at 63 out of 100 for migrants, and 71 out of 100 for natives. This indicates that respondents overestimate the extent of discrimination that migrant parents face from child care center managers when it comes to email responses.

Importantly, the cumulative distribution functions in Fig. 1(b) reveal that beliefs vary substantially across respondents. For example, the 10–90 percentile range for beliefs about migrants' child care enrollment rate spans from 10% to 79%. Beliefs about the response rate are even more dispersed, with a 10–90 percentile range of 11% to 98%. Put differently, the documented misperceptions about migrant-native gaps in early child care are relatively large compared to misperceptions in other domains found in other studies, lying in the 60–80th percentile (see literature review by Bursztyn and Yang, 2022). Thus, respondents seem to have relatively imprecise knowledge about the true extent of migrant-native inequalities in the early child care market.



# Underestimate inequality with respect to

**Fig. 2.** Correlation of demographics with prior beliefs about inequalities. *Notes*: Figure shows marginal effects from probit estimations indicating the change in the likelihood to underestimate inequality in relation to the omitted baseline category from regression models with or without control variables. *Female*: Categorical variable taking a value of one if the respondent states to be female, two if the respondent states to be diverse, and zero if the respondent states to be male (omitted); the diverse category (*n* = 10) in the gender variable is not shown. *Migrant*: Indicator variable taking a value of one if the respondent states to be diverse (omitted). *Parent*: Indicator variable taking a value of one if the respondent thas a migration background (she or either of her parents were born outside of Germany), zero otherwise (omitted). *Parent*: Indicator variable taking a value of one if the respondent is a parent (has at least one child under the age of 18 in the household), zero otherwise (omitted). *Age*: Categorical variable taking a value of one if the respondent is between 40 and 59 years old, and a value of zero if the respondent is between 18 and 39 years old (omitted). *Education*: Categorical variable taking a value of two if the respondent has completed "Higher education" (college entrance qualification, "Abitur"), a value of one if the respondent has completed "Medium education" (middle-tier secondary education ("Realschulabschluss")), and a value of zero if the respondent has completed lower education (drop out, still in school, or lower-tier secondary education ("Hauptschulabschluss")) (omitted). *Right-wing voter*: Indicator variable taking a value one if the respondent state to vote for a right-wing party (AfD, NPD, Dritter Weg, or Die Rechte), zero otherwise (omitted). *Property owner*: Indicator variable taking a value one if the respondent owns a house, zero otherwise (omitted). Error bars indicate 95% confidence intervals. See Table 2 for coefficients in the multivariate regressio

Next, we study how prior beliefs vary across respondents' sociodemographic characteristics. Specifically, we compare (i) females to males; (ii) migrants to natives; (iii) parents to non-parents; (iv) older to younger respondents; (v) those with higher educational degrees to those with lower degrees; (vi) right-wing voters to respondents with other political preferences; and (vii) property owners to non-owners (as a proxy for wealth). Fig. 2 depicts the respective subgroup coefficients when regressing enrollment rate or response rate beliefs (or both) on the subgroup indicators; the respective regression results are provided in Table 2. Specifically, the outcome variables are indicators of underestimating migrant-native gaps in the child care market with respect to enrollment rate (left panel), response rate (middle panel), or both enrollment rate and response rate (right panel). Recall that underestimating migrant-native gaps is equivalent to overestimating the values for migrants.

We find that females are 3.4 percentage points more likely to overestimate the enrollment rate of migrant children compared to males (p = .013).<sup>9</sup> While males already overestimate the enrollment rate of migrants (33.7% versus the actual value of 12%), females' enrollment beliefs are even more biased. Interestingly, migrants do not hold significantly different priors compared to natives. Parents' beliefs about the enrollment rate do not differ from those of non-parents, but parents are 4 percentage points more likely to overestimate the response rate from child care centers to migrants (p = .006).

Compared to younger respondents, those aged between 40 and 59 years and those aged 60 years and older are substantially more likely to overestimate the response rate to migrant parents (by 13.2 percentage points and 7.6 percentage points, respectively; p < .001 for both age groups). Furthermore, the degree of overestimating migrants' child care enrollment decreases in the education level. Both medium-educated respondents (by 4.2 percentage points (p = .013)) and higher-educated respondents (by 9.4 percentage points (p < .001)) are significantly less likely to overestimate enrollment rates of migrants than those with the lowest education level. Right-wing voters are 13.2 percentage points more likely to overestimate the response rate to migrant parents (p < .001), and

<sup>&</sup>lt;sup>9</sup> Note that coefficients and p-values refer to the multivariate specifications.

**T-11** 0

Table 2
Correlation between demographics and underestimation of inequality.
TT-d

	Underestimate inequality with respect to						
	Enrollment rate	Response rate	Enrollment & response				
	beliefs (1)	beliefs (2)	rate beliefs (3)				
Female	0.034**	-0.018	-0.004				
	(0.014)	(0.016)	(0.014)				
Migrant	0.033*	-0.013	0.000				
	(0.018)	(0.022)	(0.020)				
Parent	-0.008	0.045***	0.015				
	(0.014)	(0.016)	(0.015)				
40-59 years	-0.024	0.132***	0.104***				
	(0.017)	(0.020)	(0.018)				
At least 60 years	0.004	0.076***	0.050***				
	(0.018)	(0.022)	(0.019)				
Medium education	-0.042**	0.002	-0.003				
	(0.017)	(0.021)	(0.019)				
Higher education	-0.094***	-0.013	-0.021				
	(0.017)	(0.021)	(0.019)				
Right-wing voter	0.024	0.132***	0.101***				
	(0.021)	(0.027)	(0.025)				
Property owner	-0.035**	0.045***	0.030*				
	(0.014)	(0.017)	(0.016)				
Pre-specified Controls	Yes	Yes	Yes				
Ν	4570	4520	4822				

Notes: Table shows probit estimation parameters on the margin for regressions of individual characteristics on a binary indicator that takes a value of one if the respondent underestimates inequality. Results are based on multivariate probit regressions and calculated on the margin. Outcome variables are defined as follows: Column (1): Respondent underestimates inequality with regard to enrollment rates; Column (2): Respondent underestimates inequality with regard to response rates; Column (3): Respondent underestimates inequality with regard to both, zero otherwise. Female: Categorical variable taking a value of one if the respondent states to be female, two if the respondent states to be diverse, and zero if the respondent states to be male (omitted); the diverse category (n = 10) in the gender variable is not shown. Migrant: Indicator variable taking a value of one if the respondent has a migration background (she or either of her parents were born outside of Germany), zero otherwise (omitted). Parent: Indicator variable taking a value of one if the respondent is a parent (has at least one child under the age of 18 in the household), zero otherwise (omitted). Age: Categorical variable taking a value of two if the respondent is 60 years and older, a value of one if the respondent is between 40 and 59 years old, and a value of zero if the respondent is between 18 and 39 years old (omitted). Education: Categorical variable taking a value of two if the respondent has completed "Higher education" (college entrance qualification, "Abitur"), a value of one if the respondent has completed "Medium education" (middle-tier secondary education ("Realschulabschluss")), and a value of zero if the respondent has completed lower education (drop out, still in school, or lower-tier secondary education ("Hauptschulabschluss")) (omitted). Right-wing voter: Indicator variable taking a value of one if the respondent stated to vote for a right-wing party (AfD, NPD, Dritter Weg, or Die Rechte), zero otherwise (omitted). Property owner: Indicator variable taking a value one if the respondent owns a house, zero otherwise (omitted). We use survey weights to affirm national representativeness. See Appendix D for detailed variable descriptions. Robust standard errors in parentheses. Significance levels: \* p < .10, \*\* p < .05, \*\*\* p < .01.

10.1 percentage points more likely to exhibit overestimation of the combined belief measure (p. < .001). Finally, property owners are less likely to overestimate the enrollment rate of migrants (p = .015) compared to non-owners, but more likely to overestimate the response rate to migrants (p = .009). The combined measure of enrollment rate and response rate beliefs generally yields similar results as the measure of response rate beliefs alone.

In summary, our analysis reveals significant correlations between sociodemographic characteristics and prior beliefs about migrant-native gaps in early child care. Notably, respondents who are younger, more educated, and do not have right-wing voting preferences, respectively, are less likely to underestimate inequalities in the child care market.<sup>10</sup> This correlation pattern is also reflected in respondents' support for policy reforms: younger and more educated respondents show considerably higher support for reforms, whereas support is lower among right-wing voters (see Appendix Figure A1 and Appendix Table B2).<sup>11</sup>

<sup>&</sup>lt;sup>10</sup> The fact that prior beliefs vary meaningfully with respondents' characteristics (e.g., more educated respondents tend to have more accurate beliefs) suggests that the elicited priors are not mere random guesses.

 $<sup>^{11}</sup>$  A concern raised by a reviewer is that some participants might perceive the child care system itself as problematic and would therefore not support reforms — i.e., not because of the reforms themselves but because of a system viewed as flawed. However, our data suggest that this group – if it exists – is relatively small. Specifically, there are only 84 participants (1.7%) who exhibit the lowest level of support for all four proposed policies. When broadening the definition to include participants who consistently express either the lowest or second lowest level of support, or provide no response to the support questions for each of the four policies, we still identify only 257 participants (5.3%) in our dataset. Thus, our data do not support the idea that this phenomenon is of significant importance.

#### Table 3

Treatment effects on reform support.

	Centralized admission	Additional slots	Preferential treatment	Financial incentives	Reform index
	(1)	(2)	(3)	(4)	(5)
Panel A: Treatments combined					
Treatments (T1   T2   T3)	0.003 (0.049)	-0.022 (0.039)	0.044 (0.038)	-0.020 (0.048)	0.007 (0.036)
Scaled treatment effect	0.11	-0.56	2.34	-0.76	-
Control mean	3.09	3.94	1.86	2.57	-0.02
Panel B: Treatments separately					
T1: Enrollment rate	0.048	-0.003	0.081*	0.026	0.053
Information	(0.060)	(0.049)	(0.047)	(0.059)	(0.044)
T2: Response rate information	-0.032 (0.059)	-0.043 (0.049)	0.005	-0.082 (0.060)	-0.043 (0.046)
T3: Enrollment & response rate information	-0.006 (0.059)	-0.020 (0.050)	0.046 (0.046)	-0.001 (0.058)	0.012 (0.044)
Pre-specified Controls N	Yes 4634	Yes 4713	Yes 4739	Yes 4714	Yes 4767

Notes: Table shows treatment effects on an indicator for how much the respondent agreed with a given policy reform on a five-point Likert scale. Results are based on multivariate OLS regressions. Outcome variables are defined as follows: Column (1): Answer on a five-point Likert scale to the statement "Instead of individual child care center managers, a central office at the community level should decide which child gets a child care slot."; Column (2): Answer on a five-point Likert scale to the statement "The number of child care slots should be further expanded using taxpayers' money."; Column (3): Answer on a five-point Likert scale to the statement "Families with a migration background should be given preference in the allocation of child care slots."; Columns (4): Answer on a five-point Likert scale to the statement "Child care centers should receive more support from taxpayers to accommodate children with an immigrant background."; Column (5): An index combining support for all reforms. T1: Enrollment information, T2: Response rate information, and T3: Enrollment & response rate information are indicator variables taking a value of one if the respondent is in the respective treatment group, zero otherwise. Treatments (T1 | T2 | T3) is an indicator variable taking a value of one if the respondent is in any of the three treatment groups, zero otherwise. Panel A shows treatment effects for all treatment groups pooled. Panel B shows treatment effects separately for each treatment group. Reform Index is defined as follows: Applying the procedure of Kling et al. (2007), we first z-standardize the four stated policy reform support variables (Centralized admission, Increase slots, Preferential treatment, and Financial incentives) by subtracting the control-group mean and dividing by the control-group standard deviation. Then, we calculate an equally weighted average of the standardized variables and, finally, z-standardize this average. An index is computed for all respondents who have a valid response to at least one reform variable. Pre-specified Controls include gender in three categories (female, male, diverse), age in three categories (between 18 and 39 years old, between 40 and 59 years old, at least 60 years old), education in three categories (completed lower, medium, or higher education), and wealth in two categories (respondent owns property or not). We use survey weights to ensure national representativeness. See Appendix D for detailed variable descriptions. Robust standard errors in parentheses. Significance levels: \* p < .10, \*\* p < .05, \*\*\* p < .01.

# 4.2. Average treatment effects on reform support

To set the stage for analyzing information treatment effects on reform support, we first document reform support in the control group (see Table 3). Increasing the number of slots for early child care is the most popular policy reform, receiving an average support rating of 3.94 out of 5 (70% of control-group respondents "fully" or "somewhat" support this policy reform). Implementing a centralized admission system is the second most popular reform (support rating: 3.09; 40% support), followed by providing additional financial incentives for child care centers to admit migrant children (support rating: 2.57; 25% support). The least popular policy reform is granting preferential treatment of migrant children during the admission process (support rating: 1.86; 7% support).

Turning to the causal effect of providing information about migrant-native gaps on policy support, we find precisely estimated zero effects when combining all treatments (see Panel A of Table 3). Panel B confirms this finding when considering the different information treatments separately.

Although our treatments do not alter average reform support, they still significantly influence respondents' perceptions, as detailed in Appendix Table B3. Specifically, respondents who received one of the treatments are 3.8 percentage points more likely to attribute migrant-native disparities in early child care to migrants' cultural background (Column (1) of Panel A; p = .017), an increase of 13.8% relative to the control group mean. Thus, the absence of treatment effects on reform support does not imply that respondents disregard the information provided.

A likely reason for the lack of average treatment effects on reform support, despite treatment-induced shifts in perceptions about migrant-native gaps, is heterogeneity of treatment effects based on respondents' prior beliefs. Indeed, the direction in which the information treatments update beliefs should determine treatment effects on policy support (see Haaland et al., 2023). Put differently, if respondents believe that inequality is not an issue but then learn that large inequalities exist, they should increase their support for equity-enhancing policies (and vice versa). In the following section, we explore whether the overall null effect on policy support masks counterbalancing effects based on respondents' prior beliefs.

# 5. Heterogeneous treatment effects

In this section, we study heterogeneities of treatment effects on policy support in three sets of analyses. First, and most importantly, we investigate heterogeneities by prior beliefs, which will reveal whether respondents systematically differ in how



Fig. 3. Distribution of reform index for treatment and control groups by prior beliefs. *Notes*: Figure shows the unconditional distribution of the reform index for treatment and control groups by prior beliefs about inequalities in the child care market. Vertical lines report the means for treatment (solid) and control group (dashed). *Underestimator* is an indicator variable taking a value of one if the respondent answered both of the belief questions for migrants (*Prior enrollment rate, Prior response rate*) with values higher than the actual values (12 out of 100 and 63 out of 100, respectively), otherwise respondents are classified as *Overestimators*. In total, there are 1223 individuals in the control group, with 877 overestimators and 346 underestimators. Across the three treatment groups, there are 3599 individuals, with 2549 overestimators and 1050 underestimators. *Reform Index* is defined as follows: Applying the procedure of Kling et al. (2007), we first z-standardize the four stated policy reform support variables (*Centralized admission, Increase slots, Preferential treatment*, and *Financial incentives*) by subtracting the control-group mean and dividing by the control-group standard deviation. Then, we calculate an equally weighted average of the standardized variables and, finally, z-standardize this average. An index is computed for all respondents who have a valid response to at least one reform variable. We report p-values of Kolmogorov–Smirnov tests for the difference of distributions of the reform index between treatment and control groups separately for under- and overestimators in the upper two panels. In the lower panel, we report by how much the mean difference between under- and overestimators decreases in the treatment group as compared to the control group.

they react to new information given what they already know (or believe to be true) about the child care market. Second, we present an exploratory analysis of heterogeneities along different socioeconomic dimensions. Third, we present a data-driven Causal Forest analysis that identifies the primary drivers of treatment effect heterogeneities in our experimental data.

# 5.1. Heterogeneity by prior beliefs

Fig. 3 compares treatment effects on the policy support index for respondents who initially underestimated or overestimated actual migrant-native inequalities in child care. We perform the analysis on raw data (i.e., without any controls). In total, 29.0% of our sample are classified as underestimators, with the shares very similar between the control group (28.3%) and the treatment groups (29.2%). In line with expected belief updating, the treatment significantly shifts the distribution of the policy support index upward among underestimators. This can be seen by comparing the solid and dashed black lines in the upper left panel (Kolmogorov–Smirnov (K-S) test: p = .010). In the upper right panel, we find the opposite qualitative pattern for overestimators, though this effect is not statistically significant (comparing the solid and dashed gray lines; K-S test: p = .887). By combining under- and overestimators, the lower panel shows that information provision substantially reduces the polarization in reform support between both subgroups

# Table 4

Treatment effect heterogeneity by prior beliefs.

	Reform index							
	Full sample	Underestimator		Overestimator				
	(1)	(2)	(3)	(4)	(5)			
Treatments (T1   T2   T3)	-0.039 (0.042)	0.136** (0.066)		-0.036 (0.042)				
× Underestimator	0.167** (0.078)							
Underestimator	-0.322*** (0.066)							
T1: Enrollment rate information			0.242*** (0.082)		-0.014 (0.052)			
T2: Response rate information			0.053 (0.089)		-0.070 (0.053)			
T3: Enrollment & response rate information			0.123 (0.080)		-0.025 (0.051)			
Pre-specified Controls	Yes	Yes	Yes	Yes	Yes			
N	4767	1392	1392	3375	3375			

Notes: Table shows treatment effects on the Reform Index by prior beliefs. Results are based on multivariate OLS regressions. Underestimator is an indicator variable taking a value of one if the respondent answered both of the belief questions for migrants (Prior enrollment rate, Prior response rate) with values higher than the actual values (12 out of 100 and 63 out of 100, respectively), otherwise respondents are classified as Overestimators. In Column (1) we run OLS regressions on the full sample. In Columns (2)-(5), we estimate treatment effects for the subsamples of Underestimators (Columns (2) and (3)) and Overestimators (Columns (4) and (5)) separately. T1: Enrollment information, T2: Response rate information, and T3: Enrollment & response rate information are indicator variables taking a value of one if the respondent is in the respective treatment group, zero otherwise. Treatments (T1 |  $T_2 \mid T_3$  is an indicator variable taking a value of one if the respondent is in any of the three treatment groups, zero otherwise. Reform Index is defined as follows: Applying the procedure of Kling et al. (2007), we first z-standardize the four stated policy reform support variables (Centralized admission, Increase slots, Preferential treatment, and Financial incentives) by subtracting the control-group mean and dividing by the control-group standard deviation. Then, we calculate an equally weighted average of the standardized variables and, finally, z-standardize this average. An index is computed for all respondents who have a valid response to at least one reform variable. Pre-specified Controls include gender in three categories (female, male, diverse), age in three categories (between 18 and 39 years old, between 40 and 59 years old, at least 60 years old), education in three categories (completed lower, medium, or higher education), and wealth in two categories (respondent owns property or not). We use survey weights to ensure national representativeness. See Appendix D for detailed variable descriptions. Robust standard errors in parentheses. Significance levels: \* p < .10, \*\* p < .05, \*\*\* p < .01.

by 42.6% relative to the control group.<sup>12</sup> This treatment effect heterogeneity by prior beliefs explains the average null effect in the overall sample.

We confirm these graphical results in our regressions analysis accounting for the pre-specified control variables (see Table 4).<sup>13</sup> As shown in Column (1), respondents who underestimate inequalities in the child care markets are significantly less likely to support the equity-enhancing reform proposals. However, when receiving the treatment, underestimators exhibit a stronger increase in reform support than overestimators. Specifically, the treatment effect on reform support is 16.7% of a standard deviation (p = .032) higher for underestimators compared to overestimators. This pattern is also evident in Columns (2) and (4), which present treatment effects in the samples of underestimators and overestimators, respectively. While treatment effects are positive and statistically significant (p = .041) for underestimators (Column 2), they are negative, albeit insignificant (p = .392), for overestimators (Column 4).

Moreover, going beyond the combined treatment effect, Columns (3) and (5) of Table 4 present separate effects for the three information treatments. For underestimators, all treatment effects are positive, with the effect of information about the migrantnative enrollment gap (T1) being by far the largest (Column (3)). For overestimators, all three information treatments have negative effects, while none of them captures statistical significance (Columns (5)).<sup>14</sup>

Overall, these findings highlight the importance of considering respondents' prior beliefs when investigating the effects of information provision on reform support. We detect strong heterogeneous reactions to the information treatments based on

<sup>&</sup>lt;sup>12</sup> To compute the reduction in polarization, we first compare the average policy support for underestimators and overestimators in both the control and treatment groups. In the control group, the average policy support was -0.3048 for underestimators and 0.0984 for overestimators, resulting in a difference of 0.4032. In the treatment group, the average policy support was -0.1577 for underestimators and 0.0737 for overestimators, resulting in a difference of 0.2314. We then calculate the reduction in the gap as (0.4032 - 0.2314) = 0.1718, which translates to a percentage gap closure of 0.1718/0.4032 \* 100 = 42.6%.

<sup>&</sup>lt;sup>13</sup> Appendix Table B4 provides the results for each individual policy, both for the treatments combined and broken down separately by each treatment.

<sup>&</sup>lt;sup>14</sup> Results are qualitatively similar when we construct the reform index based on a principal component analysis (PCA) allowing for unequal weighting of the four policy support variables (see Appendix Table B5). If anything, results for the PCA-based reform index get slightly stronger, as the effect of the combined treatment (T3) in the subsample of underestimators becomes significant at the 10-percent-level.

respondents' priors such that information reduces polarization in respondents' preferences for equity-enhancing reforms in early child care.<sup>15</sup>

# 5.2. Heterogeneity by demographics and regional characteristics

Next, we provide an exploratory analysis of heterogeneous treatment effects among sociodemographic subgroups and regional characteristics related to child care.

*Demographics.* First, we investigate whether treatment effects are different for females (compared to males) and parents (compared to non-parents), as these two subgroups are especially impacted by child care policies owing to their active involvement in child care activities. While females and males show no significant differences in policy reform support in the control group, females react significantly more positively to the information treatment (see Column (1) of Appendix Table B6). This finding aligns with the observation that females are more likely to underestimate inequalities for enrollment rates (see Fig. 2). Parents generally show greater support for equity-enhancing policies than non-parents (see Column (2)). Notably, the information treatment leads to a significantly stronger reduction of parents' policy support.<sup>16</sup>

Second, we investigate heterogeneous treatment effects for respondents that report voting for right-wing parties. As one might expect, they exhibit less support for equity-enhancing policy reforms compared to other respondents (see Column (3) of Appendix Table B6, p < .001). Furthermore, in contrast to the predicted updating behavior that we observe in the general population, right-wing voters even *decrease* their policy support when given information about the actual migrant-native gaps in early child care (p = .001). Thus, information provision seems to reinforce, rather than mitigate, anti-migrant sentiments of these respondents. This phenomenon, wherein information exacerbates pre-existing biases, is common in situations where new information conflicts with personal beliefs (see Marino et al., 2024). In the political science literature, this is explained as an amplification of party identification based on the fact that challenging beliefs closely linked to individuals' political identity is psychologically taxing. Consequently, a typical response is to avoid this internal conflict by further aligning with the party's perspective (Campbell, 1980; Bartels, 2002).

Finally, we explored whether some respondents did not respond to the treatments because they did not believe the provided information. Although we did not directly ask respondents about their trust in the provided information, we utilized data on their trust levels towards institutions (i.e., Bundestag (German Parliament), judiciary, police, politicians, political parties, and the European Parliament), which the *Inequality Barometer* measured on a scale from 0 to 10. Regressing our reform index on the treatment indicator, respondents' trust in institutions (indexed using the standardization approach by Kling et al. (2007)), and their interaction, we find a positive and significant interaction term (see Column (4) of Appendix Table B6). At face value, this result suggests that distrust among some respondents may indeed attenuate treatment effects. However, we caution against over-interpreting this result, as our trust measure is only an indirect proxy for trust in the provided information.

*Regional characteristics.* It also seems crucial to understand regional heterogeneities in both reform support and treatment effects, as both the availability and the institutional setting of child care varies substantially across regions. Each federal state in Germany has the authority to establish and manage its own overarching child care regulations and frameworks. Administrative responsibility, however, including the responsibility to set (income-based) fees prior to the age when child care becomes free of charge, lies with each municipality. We investigated whether general support for child care reforms and the observed treatment effects vary by regional characteristics, including (i) financial incentives for child care centers to enroll migrant children, (ii) whether child care is free for certain age ranges, (iii) urbanity level, and (iv) degree of child care slot rationing. The first two variables are measured at the federal state level (k = 16), while the latter two are measured at the county level (k = 398). While the state-level variables capture the "right" variation (as financial incentives and fee exemptions vary at the state level), we would have preferred to use information on urbanity and the degree of slot rationing at the municipality level. However, such finely grained data do not exist in Germany.

Results are presented in Appendix Table B7. First, as shown in the lower part of the table, support for child-care-related policy reforms in the control group shows no systematic variation with most regional characteristics. However, support is significantly higher in the 11 states where child care is free for certain age ranges compared to the five states that charge fees until age six. This correlation could be due to, among many other things, positive experiences with existing policies, economic relief, or a generally more supportive cultural and political environment that may enhance support for further reforms. Nonetheless, these comparisons should be interpreted cautiously due to the limited number of federal states and potential unobserved state-specific factors.

Second, there is no evidence for heterogeneity of our treatment effects by regional characteristics (see upper part of Appendix Table B7). Interactions of the treatment indicator with the regional variables are not only statistically insignificant, but also economically small. This consistency in the results suggests that the absence of significant treatment effects for Germany as a whole does not mask important heterogeneities at the subnational level.

<sup>&</sup>lt;sup>15</sup> The reported heterogeneities by prior beliefs also indicate that the treatment effects on policy preferences operate through "information-based updating", as opposed to other possible channels like "salience-based updating" (Bleemer and Zafar, 2018). While the former means that effects are due to correcting biased beliefs, the latter simply means that the information treatment makes certain aspects – in our case, inequality in early child care – more salient. Heterogeneities by prior beliefs, which we find, are often interpreted as evidence for information-based updating (Lergetporer and Woessmann, 2023). This is also consistent with our experimental setup, as eliciting prior beliefs about migrant-native gaps among all respondents (including control group respondents) should mute potential salience-based effects.

<sup>&</sup>lt;sup>16</sup> Results hold for different subgroups of parents. For instance, parents of children under the age of ten also exhibit greater policy support compared to non-parents (coef. = .192, p = .032), which diminishes significantly after exposure to the treatment information (coef. = -.229, p = .032).



Scatter Plot of Response Rate Beliefs against CATEs with Fitted Lines

Fig. 4. Scatter plot of CATEs and response rate beliefs. *Notes*: Figure shows individual CATEs (y-axis) plotted against respondents' response rate belief percentiles (x-axis). CATEs are the result of a Causal Forest with 25,000 trees as described in Appendix F. The solid line is a fitted line minimizing mean squared errors. The dashed line is a quadratic fitted line minimizing mean squared errors. Vertical lines indicate the actual response rates to migrants (dotted) and natives (solid), taken from Hermes et al. (2023a).

# 5.3. Causal forest estimation

Finally, we conduct a Causal Forest analysis (Wager and Athey, 2018; Athey and Wager, 2019) for three main reasons. First and most importantly, the Causal Forest analysis is a data-driven approach that identifies the primary drivers of treatment effect heterogeneities in our sample. Second, the analysis provides valuable insights into the functional form of heterogeneity, revealing complex interactions and non-linearities. Third, the analysis addresses concerns about external validity by identifying subgroups with differing treatment effects, thereby supporting the generalizability of our findings.

Conceptually, the Causal Forest analysis divides the data into subsets along different covariates, and subsequently evaluates the treatment effect within each subset. The method calculates the Conditional Average Treatment Effects (CATEs) for each respondent by averaging these effects across numerous trees, accounting for confounding variables. It also determines the importance of different variables for driving treatment effect heterogeneities by gauging their contribution to predictive accuracy (see Appendix F for details).

In our application, using all covariates collected in the survey, the Causal Forest highlights prior beliefs about migrants' enrollment and response rates as major drivers of treatment effect heterogeneity. In Fig. 4, we illustrate the relationship between each respondent's belief about the response rate to migrant inquiries and their individual CATEs (see Appendix Figure F1 for the analogous plot for enrollment rate beliefs). The graph shows significant variation in treatment effects based on these prior beliefs. Respondents who perceive no migrant-native gap in the response rate or who believe migrants receive more responses than natives typically increase their reform support upon receiving treatment information. Conversely, those who perceive little discrimination (with response rate beliefs ranging from about 30 to 60) show minimal to no change in reform support. However, respondents who greatly underestimate the response rate (thus overestimating inequalities) tend to reduce their support for policy reforms after receiving the information. These trends are further supported by additional analyses, including quadratic OLS and quartile regressions, as detailed in Appendix Table B8.

In sum, the Causal Forest analysis highlights prior beliefs about migrant-native inequalities as main drivers of treatment effect heterogeneity. This finding is reassuring, as it echoes our conceptual considerations which led us to focus our main heterogeneity analysis on these beliefs (see Section 5.1).

# 6. Discussion of possible limitations

Having presented our empirical results, we now discuss some potential limitations of our study and suggest directions for future research.

First, the fact that our main outcomes are survey-based stated preferences, without direct economic incentives, merits further discussion. While frequently used in contemporary economic research to assess people's (policy) preferences (Falk et al., 2018; Haaland et al., 2023), these measures are occasionally met with skepticism due to their possible susceptibility to various types of reporting bias. For instance, survey-based preferences, lacking immediate political or economic consequences, may be susceptible to *hypothetical bias*. This occurs when respondents' answers, based on hypothetical scenarios, do not reflect their actual behaviors in real-life situations. This bias is particularly noted in surveys valuing non-market goods, such as pristine natural environments (Carson, 2012; Kling et al., 2012). Furthermore, survey responses on sensitive issues can be affected by *social desirability bias*. For example, Coffman et al. (2017) demonstrate that the reported prevalence of antigay sentiments in the U.S. population is understated in standard surveys. Finally, Falk and Zimmermann (2013) identify *preferences for consistency* as another potential source of bias in survey-reported preferences. Specifically, they find that including certain questions in surveys influences respondents' answers to subsequent related questions. For example, asking respondents if everyone deserves a second chance can reduce their agreement to lifetime imprisonment for murderers in subsequent questions.

Despite potential biases in survey-based stated preference measures, it is reassuring that multiple pieces of evidence indicate the relevance of survey responses in actual political processes. For instance, Hainmueller et al. (2015) highlight the external validity of survey experiments, showing that the outcomes of hypothetical survey experiments match with the results from similar real-world referendums on immigration policies (see also Alesina et al., 2023; Haaland and Roth, 2023; Lergetporer and Woessmann, 2023, for further evidence). Similarly, survey-based preference measures often correlate strongly with incentivized outcomes such as donations or signing a petition (Haaland et al., 2023). More generally, Blinder and Krueger (2004) suggest that public-opinion surveys are politically significant, as evidenced by the substantial resources politicians allocate to polling to inform their policymaking. Several empirical studies support this view. For instance, Hager and Hilbig (2020) provide quasi-experimental evidence that German politicians' policy stances are indeed shaped by public opinions as reflected in surveys. In a similar vein, Butler and Nickerson (2011) randomly sent letters with district-specific survey results on policy proposals to legislators and find that this information increased the likelihood of voting in alignment with constituent opinions. Similarly, Liagat (2020) demonstrates that informing politicians about citizens' preferences on various policy issues increases the likelihood that they will recommend policies aligning with those preferences to their party leadership. Additionally, Banerjee et al. (2024) experimentally show that Delhi municipal councillors who received a pre-election report card two years prior to elections increased pro-poor spending in high-slum areas and influenced electoral outcomes. In summary, although we recognize the potential limitations of survey-based preference measures, the evidence discussed gives us confidence that our outcomes accurately reflect respondents' policy preferences. For future research, exploring the direct effects of our information treatments on political actions, such as signing petitions or voting behavior, would be insightful.

Second, the reasons for why effects on reform support differ across treatments require additional explanation. In particular, one intriguing result in Table 4 is that effects for T1 are considerably larger than for T3, which contains some of the same information (i.e., information about the migrant-native enrollment gap). One potential explanation for the difference in the magnitude of the coefficients is information overload (e.g. Bawden and Robinson, 2020). T1 offers a single, clear piece of information, making it easier for respondents to comprehend and use as a reference point. In contrast, T3 presents two pieces of information, which may overwhelm respondents and thus inhibit effectiveness. Additionally, the first piece of information in T3 might act as a benchmark for the second, potentially causing cognitive dissonance or confusion (for supporting evidence, see Mierisch, 2024). Furthermore, the difference might be due to the specific nature of the information; enrollment rates (T1) are more tangible and easily understood, whereas response rates (part of T3) are more abstract. While we can only speculate on these explanations, it would be interesting for future work to investigate the underlying reasons for this difference in more detail.

# 7. Conclusions

In this paper, we present results from a representative survey experiment investigating how information about migrantnative gaps in access to early child care affects public preferences for equity-enhancing policy reforms. Respondents have strong misperceptions about inequalities in early child care, overestimating the enrollment rate of migrants and underestimating the response rate of child care centers to inquiries from migrant parents. While providing factual information about the extent of migrant-native gaps successfully updates respondents' beliefs, it has no average effect on reform support.

Importantly, the overall null effect of information provision on reform support masks two countervailing effects for respondents with different prior beliefs about migrant-native gaps: respondents who initially underestimated inequalities increase their reform support upon receiving the information. On the other hand, those who initially overestimated inequalities tend to decrease reform support, albeit not statistically significantly so. Put together, correcting misperceptions through providing factual information narrows the gap in reform support between these two groups by as much as 43%, suggesting that information provision can reduce polarization in policy preferences.

Finally, we would like to highlight some policy implications of our results. First, we elucidate the degree of public support of different equity-enhancing policy reforms in early child care. We find that providing additional child care slots is particularly well received by the public, even enjoying majority appeal. Second, our results show a lack of prior knowledge (or awareness) about the early child care market, as indicated by the considerable variation in prior beliefs about enrollment and response rates. This lack of knowledge could explain why we find that providing factual information decreases polarization in reform support. In contrast, in settings with a better informed population, studies tend to find that information provision either increases polarization in reform support or does not affect polarization at all (see literature review by Marino et al., 2024). Our findings suggest that informational campaigns could aid policymakers in achieving greater consensus on policies targeting societal inequality in less-debated topics.

# Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

#### Appendix A. Supplementary material

Supplementary material related to this article can be found online at https://doi.org/10.1016/j.jebo.2024.106780.

#### Data availability

Data will be made available on request.

#### References

- Alesina, A., Miano, A., Stantcheva, S., 2023. Immigration and redistribution. Rev. Econ. Stud. 90 (1), 1-39.
- Almond, D., Currie, J., 2011. Human capital development before age five. Handbook of labor economics. Handb. Labor Econ. Ed. 1, 1315-1486.
- Athey, S., Wager, S., 2019. Estimating treatment effects with causal forests: An application. Observ. Stud. 5 (2), 37-51.
- Banerjee, A., Enevoldsen, N., Pande, R., Walton, M., 2024. Public information is an incentive for politicians: Experimental evidence from Delhi elections. Am. Econ. J.: Appl. Econ. 16 (3), 323–353.
- Bartels, L.M., 2002. Beyond the running tally: Partisan bias in political perceptions. Political Behav. 24, 117-150.
- Bawden, D., Robinson, L., 2020. Information overload: An introduction. In: Oxford Research Encyclopedia of Politics.
- Bleemer, Z., Zafar, B., 2018. Intended college attendance: Evidence from an experiment on college returns and costs. J. Public Econ. 157 (1), 184-211.
- Blinder, A.S., Krueger, A.B., 2004. What does the Public Know About Economic Policy, and How Does It Know It? Working Paper w10787, In: National Bureau of Economic Research, National Bureau of Economic Research.
- Bonomi, G., Gennaioli, N., Tabellini, G., 2021. Identity, beliefs, and political conflict. Q. J. Econ. 136 (4), 2371-2411.
- Bundesamt für Migration, 2019. Migrationsbericht der Bundesregierung Migrationsbericht 2019. Bundesminsterium des Inneren.
- Bursztyn, L., Yang, D.Y., 2022. Misperceptions about others. Annu. Rev. Econ. 14, 425–452.
- Butler, D.M., Nickerson, D.W., 2011. Can learning constituency opinion affect how legislators vote? Results from a field experiment. Q. J. Political Sci. 6 (1), 55-83.
- Campbell, A., 1980. The American Voter. University of Chicago Press.
- Campbell, F., Conti, G., Heckman, J.J., Moon, S.H., Pinto, R., Pungello, E., Pan, Y., 2014. Early childhood investments substantially boost adult health. Science 343 (6178), 1478-1485.
- Carlsson, S., Thomsen, S.L., 2015. Improving the Allocation of Spots in Child Care Facilities for Toddlers in Germany: A Mechanism Design Approach. Working paper, In: IZA Discussion Paper, IZA Discussion Paper.
- Carson, R.T., 2012. Contingent valuation: A practical alternative when prices aren't available. J. Econ. Perspect. 26 (4), 27-42.
- Cattaneo, M., Lergetporer, P., Schwerdt, G., Werner, K., Woessmann, L., Wolter, S.C., 2020. Information provision and preferences for education spending: Evidence from representative survey experiments in three countries. Eur. J. Political Econ. 63, 101876.
- Coffman, K.B., Coffman, L.C., Ericson, K.M.M., 2017. The size of the LGBT population and the magnitude of antigay sentiment are substantially underestimated. Manage. Sci. 63 (10), 3147–3529.
- Corak, M., 2013. Income inequality, equality of opportunity, and intergenerational mobility. J. Econ. Perspect. 27 (3), 79-102.
- Dixit, A.K., Weibull, J.W., 2007. Political polarization. Proc. Natl. Acad. Sci. 104 (18), 7351-7356.
- DJI, 2021. Fachkräftebarometer Frühe Bildung 2021. Weiterbildungsinitiative Frühpädagogische Fachkräfte, Autorengruppe Fachkräftebarometer, München.

Drange, N., Havnes, T., 2019. Early childcare and cognitive development: Evidence from an assignment lottery. J. Labor Econ. 37 (2), 581-620.

- Education Report, 2020. Autorengruppe Bildungsberichterstattung: Bildung in Deutschland 2020. Einindikatorengestützter Bericht mit einer Analyse zur Bildung in einer digitalisierten Welt, Bundesministerium für Bildung und Forschung.
- Falk, A., Becker, A., Dohmen, T., Enke, B., Huffman, D., Sunde, U., 2018. Global evidence on economic preferences. Q. J. Econ. 133 (4), 1645–1692.
- Falk, A., Zimmermann, F., 2013. A taste for consistency and survey response behavior. CESifo Econ. Stud. 59 (1), 181-193.
- Felfe, C., Lalive, R., 2018. Does early child care affect children's development? J. Public Econ. 159, 33-53.
- Fugger, N., Klein, T., Riehm, T., 2017. Dezentrale kitaplatzvergabe ohne warteschlange: Ein leitfaden. ZEW Policy Brief 2017.
- García, J.L., Heckman, J.J., Leaf, D.E., Prados, M.J., 2020. Quantifying the life-cycle benefits of an influential early-childhood program. J. Polit. Econ. 128 (7), 2502–2541.
- Grewenig, E., Lergetporer, P., Werner, K., 2020. Gender Norms and Labor-Supply Expectations: Experimental Evidence from Adolescents. CESifo Working Paper No. 8611, In: Working Paper, Center for Economic Studies and ifo Institute.
- Grigorieff, A., Roth, C., Ubfal, D., 2020. Does information change attitudes toward immigrants? Demography 57 (3), 1117-1143.
- Guriev, S., Papaioannou, E., 2022. The political economy of populism. J. Econ. Lit. 60 (3), 753-832.
- Haaland, I., Roth, C., 2023. Beliefs about racial discrimination and support for pro-black policies. Rev. Econ. Stat. 105 (1), 40-53.
- Haaland, I., Roth, C., Wohlfart, J., 2023. Designing information provision experiments. J. Econ. Lit. 61 (1), 3-40.
- Hager, A., Hilbig, H., 2020. Does public opinion affect political speech? Am. J. Political Sci. 64 (4), 921-937.
- Hainmueller, J., Hangartner, D., Yamamoto, T., 2015. Validating vignette and conjoint survey experiments against real-world behavior. Proc. Natl. Acad. Sci. 112 (8), 2395–2400.
- Heckman, J.J., Landersø, R., 2021. Lessons from Denmark about Inequality and Social Mobility. Working Paper w28543, In: National Bureau of Economic Research, National Bureau of Economic Research.
- Heckman, J.J., Moon, S.H., Pinto, R., Savelyev, P.A., Yavitz, A., 2010. The rate of return to the high scope perry preschool program. J. Public Econ. 94 (1-2), 114-128.
- Heckman, J., Pinto, R., Savelyev, P., 2013. Understanding the mechanisms through which an influential early childhood program boosted adult outcomes. Amer. Econ. Rev. 103 (6), 2052–2086.
- Hermes, H., Lergetporer, P., Mierisch, F., Peter, F., Wiederhold, S., 2023a. Discrimination in Universal Social Programs? A Nationwide Field Experiment on Access to Child Care. CESifo Working Paper No. 10368, In: Working Paper, Center for Economic Studies and ifo Institute.
- Hermes, H., Lergetporer, P., Mierisch, F., Peter, F., Wiederhold, S., 2023b. Males should mail? Gender discrimination in access to childcare. AEA Pap. Proc. 113, 427–431.

Hermes, H., Lergetporer, P., Peter, F., Wiederhold, S., 2024. Application Barriers and the Socioeconomic Gap in Child Care Enrollment. Forthcoming at the Journal of the European Economic Association.

Hopkins, D.J., Sides, J., Citrin, J., 2019. The muted consequences of correct information about immigration. J. Politics 81 (1), 315-320.

Jessen, J., Schmitz, S., Waights, S., 2020. Understanding day care enrolment gaps. J. Public Econ. 190, 104252.

Karakas, L.D., Mitra, D., 2019. Immigration Policy in the Presence of Identity Politics. Working Paper.

Kling, J.R., Liebman, J.B., Katz, L.F., 2007. Experimental analysis of neighborhood effects. Econometrica 75 (1), 83-119.

Kling, C.L., Phaneuf, D.J., Zhao, J., 2012. From exxon to BP: Has some number become better than no number? J. Econ. Perspect. 26 (4), 3-26.

Lergetporer, P., Piopiunik, M., Simon, L., 2021. Does the education level of refugees affect natives' attitudes? Eur. Econ. Rev. 134, 103710.

Lergetporer, P., Schwerdt, G., Werner, K., West, M.R., Woessmann, L., 2018. How information affects support for education spending: Evidence from survey experiments in Germany and the United States. J. Public Econ. 167, 138–157.

Lergetporer, P., Woessmann, L., 2023. Earnings information and public preferences for university tuition: Evidence from representative experiments. J. Public Econ. 226, 104968.

Liaqat, A., 2020. No Representation without Information: Politician Responsiveness to Citizen Preferences. Working paper.

Marino, M., Iacono, R., Mollerstrom, J., 2024. (Mis-) perceptions, information, and political polarization: a survey and a systematic literature review. European Journal of Political Economy 102578.

Mierisch, F., 2024. Does awareness of achievement gaps influence concerns about multiple dimensions of educational inequality? mimeo.

Naumann, E., F. Stoetzer, L., Pietrantuono, G., 2018. Attitudes towards highly skilled and low-skilled immigration in europe: A survey experiment in 15 European countries. Eur. J. Political Res. 57 (4), 1009–1030.

OECD, 2018. Settling In 2018. Technical report, OECD, Paris.

Roth, C., Settele, S., Wohlfart, J., 2022. Beliefs about public debt and the demand for government spending. J. Econometrics 231 (1), 165-187.

Settele, S., 2022. How do beliefs about the gender wage gap affect the demand for public policy? Am. Econ. J.: Econ. Policy 14 (2), 475-508.

Spiegel, 2010. Volksentscheid: Hamburger schmettern Schulreform ab. URL https://www.spiegel.de/lebenundlernen/schule/volksentscheid-hamburger-schmetternschulreform-ab-a-707179.html. (Accessed July 2024).

Spiess, C.K., 2013. Investments in education: The early years offer great potential. DIW Econ. Bull. 3 (10), 3-10.

SZ, 2024. München: Protest gegen Fördersystem für Kitas. URL https://www.sueddeutsche.de/muenchen/muenchen-kitas-gebuehren-protest-1.6551215. (Accessed July 2024).

Wager, S., Athey, S., 2018. Estimation and inference of heterogeneous treatment effects using random forests. J. Amer. Statist. Assoc. 113 (523), 1228–1242.